Author’s response to reviews

Title: Open reduction and internal fixation of humeral midshaft fractures: anterior versus posterior plate fixation

Authors:
Sebastian Lotzien (sebastianlotzien@hotmail.com;sebastian.lotzien@bergmannsheil.de)
Clemens Hoberg (Clemens.Hoberg@bergmannsheil.de)
Valentin Rausch (Valentin.rausch@bergmannsheil.de)
Thomas Rosteius (Thomas.Rosteius@bergmannsheil.de)
Thomas Schildhauer (Thomas.schildhauer@bergmannsheil.de)
Jan Gessmann (Jan.Gessmann@bergmannsheil.de)

Version: 2 Date: 02 Oct 2019

Author’s response to reviews:

Manuscript number: BMSD-D-19-00728

Open reduction and internal fixation of humeral midshaft fractures: anterior versus posterior plate fixation
Sebastian Lotzien, MD; Clemens Hoberg, MD; Valentin Rausch, MD; Thomas Rosteius, MD; Thomas Armin Schildhauer, PhD; Jan Gessmann, PhD

Dear reviewers,

Thank you very much for the review of the manuscript.

Please find our answers to your comments and questions below.

Reviewer reports:

Robinson Esteves Pires, M.D, Ph.D (Reviewer 1):
Just one question: I was unable to find tables 1/2 and figures 1/2 in the revision 1 file.
Best regards,
Robinson Pires

Dear Reviewer, thank you for the helpful responses. I just uploaded the manuscript including the tables and figures now.
Best regards
Sebastian Lotzien

Christian von Rüden, M.D. (Reviewer 2):
the paper is almost ready for publication now. Tables 1 and 2 as well as Figures 1.. and 2... were missing in your revision. Please, provide them and resubmit the manuscript.

Dear Reviewer, thank you for the helpful review process. I do think the review process was very constructive and helped to rise the quality of the manuscript. I just uploaded the manuscript including the tables and figures now.

Best regards
Sebastian Lotzien